\listfiles
\documentclass[dvipdfmx]{jarticle}
\usepackage{amsmath}

\usepackage{graphicx}
%\usepackage{atbegshi}
%\AtBeginShipoutFirst{\special{pdf:tounicode 90ms-RKSJ-UCS2}}
%\AtBeginDvi{\special{pdf:tounicode EUC-UCS2}}

% \AtBeginDvi{\special{pdf:tounicode 90ms-RKSJ-UCS2}} % 2007-06-14 6.76i以前

\usepackage[setpagesize=false,dvipdfmx,%
bookmarks=true,bookmarksnumbered=true,%
pdftitle={タイトル},%
pdfauthor={作成者},%
pdfsubject={サブタイトル},%
pdfkeywords={キーワード},
linkcolor=blue,anchorcolor=blue,urlcolor=blue,
]{hyperref}
\usepackage{pxjahyper}

\usepackage{fancyhdr,lastpage}

\fancypagestyle{head}
{%
\fancyhf{} 
\cfoot{\thepage{}/{}\protect\pageref*{LastPage}}
\renewcommand{\headrulewidth}{0.0pt}
}

\begin{document}

\pagestyle{head}

\begin{center}%
{\LARGE タイトル \par}%
\vskip 3em%
{\Large
 \lineskip .75em%
\begin{tabular}[t]{c}%
著者
\end{tabular}\par}%
\vskip 1.5em%
{\large \today \par}% 
\end{center}\par

\section{A}

\href{http://www.apple.com/}{Apple}

\href{http://www.apple.com/}{http://www.apple.com/}

\subsection{aaa}

\begin{equation*}\label{eq}
a=1
\end{equation*}

\eqref{eq}です。

\cite{manual}、\cite{texbook}参照。

この文章は全部で\pageref{lastp}ページあります。

\newpage

Last Page

\begin{thebibliography}{99}
\bibitem{manual} Leslie Lamport.\newblock \emph{{\LaTeX:} A Document
Preparation System}.\newblock Addison-Wesley, Reading,
Massachusetts, second edition, 1994, ISBN~0-201-52983-1.

\bibitem{texbook} Donald~E. Knuth.\newblock \textit{The \TeX{}book,}
Volume~A of \textit{Computers and Typesetting}, Addison-Wesley,
Reading, Massachusetts, second edition, 1984, ISBN~0-201-13448-9.

\end{thebibliography}
\label{lastp}

\end{document}
